Your Name: _______________________________________

Quiz 2 Page 9 of 8 CS 150 - Sp. 96

(4) (b) Obtain a race-free state assignment for the merged

flow table shown on the right ( not related to part 4(a))

using as few additional states as possible . List all

adjacency constraints. Show all steps and your final state-

codes. (Hint: three state variables are required.)

4. (b) (10pts)

Additional space for Problem 4(b) x 1 x 2

00 01 11 10

1 2 3 4

7 6 3 5

7 6 8 4

7 2 3 5

CS 150 - Sp. 96 Page 8 of 8 Quiz 2